Author's response to reviews

Title: A new molecular breast cancer subclass defined from a large scale real-time quantitative RT-PCR study

Authors:

Maia Chanrion (mboulfroy@valdorel.fnclcc.fr)
Helene Fontaine (hfontaine@valdorel.fnclcc.fr)
Carmen Rodriguez (crodriguez@valdorel.fnclcc.fr)
Vincent Negre (vnegre@valdorel.fnclcc.fr)
Frederic Bibeau (fbibeau@valdorel.fnclcc.fr)
Charles Theillet (theillet@valdorel.fnclcc.fr)
Alain Henaut (alainhenaut@yahoo.fr)
Jean-Marie Darbon (jmdarbon@valdorel.fnclcc.fr)

Version: 4 Date: 2 March 2007

Author's response to reviews:

Please find enclosed the final version of our manuscript that we have revised according to the Editorial Board recommendations.
Sincerely yours